Author's response to reviews

Title: The MacGyver Effect: Alive and well in health services research?

Authors:

Roshan Perera (roshan.perera@otago.ac.nz)
Helen J Moriarty (helen.moriarty@otago.ac.nz)

Version: 3 Date: 10 September 2011

Author's response to reviews: see over
10 September 2011

Dear Sir/Madam

On behalf of my colleague and myself, I would like to re-submit the revised manuscript with requested formatting changes, of the debate article “The MacGyver Effect: Alive and well in health services research?”.

Requested formatting changes have been made as itemised below.

1. Author contributions: An additional sentence has been added to indicate that both authors have read and approved the final manuscript.

2. References are numbered consecutively, in the order in which they are cited in the text. References 15 and 16 have been cited within the main text. The numerical order has been maintained.

3. Tracking has been removed from the manuscript.

4. The manuscript has been checked for typographical errors.

With thanks

Yours truly

Roshan Perera